Author’s response to reviews

Title: Cardiac valve calcification and risk of cardiovascular or all-cause mortality in dialysis patients: A meta-analysis

Authors:

Zhe Wang (cmu_wz@hotmail.com)
Aili Jiang (steelteam2001@163.com)
Fang Wei (weiwei225@sina.com)
Haiyan Chen (haiych0716@126.com)

Version: 6 Date: 14 Jan 2018

Author’s response to reviews:

BCAR-D-17-00051R5

Cardiac valve calcification and risk of cardiovascular or all-cause mortality in dialysis patients: A meta-analysis

Zhe Wang; Aili Jiang; Fang Wei; Haiyan Chen

BMC Cardiovascular Disorders

1. Please provide further details regarding the author contributions of WZ, JAL, CHY, and WF, as currently these authors do not automatically qualify for authorship.

According to the ICMJE guidelines, and author should have:

a) made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data; AND

b) been involved in drafting the manuscript or revising it critically for important intellectual content; AND

c) given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content; AND

d) agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.
Anyone listed as an author must be included in this section. If you choose to change your author list you will need to fill out a change in authorship form and send it by email to the Editorial office to be approved by the Editor. The form can be found here: https://www.biomedcentral.com/getpublished/editorial-policies#authorship.

Anyone who contributed towards the article who does not meet the criteria for authorship can be acknowledged in the ‘Acknowledgements’ section.

Reply: We have revised the manuscript according to your comments, it can be found in ‘Authors' contributions’ section, line 3-7, page 12).

2. We note that your paper is a meta-analysis however you have included the following statement ‘The study protocols were approved by the Ethical Committee of The Second Hospital of Tianjin Medical University. The content for publication was obtained from the person participated in this study.’ Please clarify why this has been included as your study does not appear to contain any participants.

Reply: We have revised the manuscript according to your comments, it can be found in ‘Ethics approval and consent to participate’ section, line 16, page 11).

3. We also note that you have provided the following statement within the ‘Availability of data’ section ‘The agreement signed with the Ministry of Health, granting the authors access to the data bases, imposes restrictions forbidding access to them by other research groups or for other purposes’. Please can you confirm which data bases were used which require permission?

Reply: We have revised the manuscript according to your comments, it can be found in ‘Availability of data and material’ section, line 20, page 11).

4. Please include the email addresses for all authors on the title page. The corresponding author should still be indicated.

Reply: We have revised the manuscript according to your comments, it can be found in ‘E-mail’ section, line 11-15, page 11).

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Reply: We have revised the manuscript according to your comments.